Author's response to reviews

Title: Our clinical experiences and success rates of acromegaly treatment: The single center results of 62 patients

Authors:

Mehtap Evran (mehtap.evran@hotmail.com)
Murat Sert (muratser@cu.edu.tr)
Tamer Tetiker (tamertet@cu.edu.tr)

Version: 3 Date: 4 August 2014

Author's response to reviews: see over
To,
The Editors,

Subject: Submission of Original article titled, “Our clinical experiences and success rates of acromegaly treatment: The single center results of 62 patients”

Dear Sir/Ma’am

We are submitting an original article titled, “Our clinical experiences and success rates of acromegaly treatment: The single center results of 62 patients” for consideration for publication in BMC Endocrine Disorders. This clinical work not includes new information about treatments of acromegaly, but we think that it can remind us of the problems in acromegalic patients of the following: Low success rates of treatments, the important of lack of experienced surgeons and the problems in follow-up. We hope that the article will provide contributions to the literature. We hope it will be convenient for the journal, of course any suggestion is welcome. We look forward to hearing from you.

We, Mehtap Evran, Murat Sert and Tamer Tetiker certify that:

- The manuscript is being submitted by me (Mehtap Evran) on behalf of all the authors.
- The manuscript is original work of all authors.
- All authors made a significant contribution to this study.
- This manuscript has not been submitted for publication; it has not been accepted for publication and has not been published in any other journal.
- All authors have read and approved the final version of the manuscript.

Thank you

Sincerely,

The author details:
Mehtap Evran, M.D. Assist. Prof. Dr. Specialist in Internal Medicine and Endocrinology, Cukurova University, Adana, Turkey, mehtap.evran@hotmail.com

Murat Sert, M.D. Prof. Dr. Specialist in Internal Medicine and Endocrinology, Cukurova University, Adana, Turkey, muratser@cu.edu.tr

Tamer Tetiker, M.D. Prof. Dr. Specialist in Internal Medicine and Endocrinology, Cukurova University, Adana, Turkey, tamertet@cu.edu.tr

CORRESPONDING AUTHOR DETAILS
Mehtap Evran,
mehtap.evran@hotmail.com
Cukurova University Medical Faculty, Dep. of Internal Medicine, Div. of Endocrinology, Balcali Hospital, 01330, Adana, Turkey
Contact Phone Number: + 905327818634
Fax: 00903223386879
Contact Email: mehtap.evran@hotmail.com
Responses to Reviewers

The manuscript was revised as following;

1. Word count of manuscript, number of pages and number of lines were added to title page section.

2. In line 118, we reported name of ethics committee.

3. Line numbers were added to all pages of the manuscript

Sincerely,
05.08. 2014

Mehtap Evran,
mehtap.evran@hotmail.com
Cukurova University Medical Faculty, Dep. of Internal Medicine, Div. of Endocrinogy, Balcali Hospital, 01330, Adana, Turkey
Contact Phone Number: + 905327818634
Fax: 00903223386879